Author’s response to reviews

Title: Terminal dribbling in male patients with lower urinary tract symptoms: Relationship with International Prostate Symptom Score and intravesical prostatic protrusion

Authors:

Jae Heon Kim (piacekjh@hanmail.net)
Ji Sung Shim (colllaterall@gmail.com)
Hoon Choi (doc71377@gmail.com)
Du Geon Moon (dgmoon@korea.ac.kr)
Jeong Gu Lee (jeongkl@kumc.or.kr)
Je Jong Kim (jjkim98@korea.ac.kr)
Jae Hyun Bae (urobae@genetherapy.or.kr)
Jae Young Park (jaeyoungpark@korea.ac.kr)

Version: 3
Date: 28 July 2015

Author’s response to reviews: see over
Author's response to reviews

Title: Terminal dribbling in male patients with lower urinary tract symptoms: Relationship with International Prostate Symptom Score and intravesical prostatic protrusion

Authors:
Jae Heon Kim (piacekjh@hanmail.net)
Ji Sung Shim (collateralli@gmail.com)
Hoon Choi (doc71377@gmail.com)
Du Geon Moon (dgmoon@korea.ac.kr)
Jeong Gu Lee (jeongkl@kumc.or.kr)
Je Jong Kim (jjkim98@korea.ac.kr)
Jae Hyun Bae (urobae@genetherapy.or.kr)
Jae Young Park (jaeyoungpark@korea.ac.kr)

Version: 2 Date: 31 May 2015

Author's response to reviews: see over
EDITORIAL REQUEST:

1. Format Abstract according to our guidelines

2. Remove the Ethical standard section and include the information and name of ethics committee in the Methods section

→ We followed all of editorial request

Reviewer's report

Title: Terminal dribbling in male patients with lower urinary tract symptoms: Relationship with International Prostate Symptom Score and intravesical prostatic protrusion

Version: 2  Date: 11 June 2015

Reviewer: Fabrizio Presicce

Reviewer's report:

The authors have adequately resolved the limitations of the previous revision

Level of interest: An article whose findings are important to those with closely related research interests

Quality of written English: Acceptable

Statistical review: Yes, and I have assessed the statistics in my report.

Declaration of competing interests:

I declare that I have no competing interests